Author’s response to reviews

Title: Irrational use of antimalarial drugs in rural areas of eastern Pakistan: A random field study

Authors:

Shafaat Y Khan (shafaatyarkhan@hotmail.com)
Ahmad Khan (aashirahmed_86@yahoo.com)
Hafiz M Tahir (hafiztahirpk1@yahoo.com)
Muhammad Arshad (arshad_pk2002@yahoo.com)
Muhammad K Mukhtar (mkmukhtar@gmail.com)
Khawaja R Ahmad (k.r.ahmad@gmail.com)
Najma Arshad (najmaarshad@yahoo.com)

Version: 6 Date: 5 October 2012

Author’s response to reviews:

Dear Editor,

I am submitting the revised manuscript, after getting it edited by the Edanz as per recommendation of editorial team. I hope this will be fine this time.

Regards
Shafaat Yar Khan
Lecturer
Department of Biological Sciences
University of Sargodha, Sargodha
Pakistan